Author's response to reviews

Title: Are modern contraceptives acceptable to people and where do they source them from across Nigeria?

Authors:

Jane C Enemuoh (chinelo.enemuo@yahoo.com)
Obinna E Onwujekwe (onwujekwe@yahoo.co.uk)
Chinyere Mbachu (chinyere23ng@yahoo.com)
Chinwe Ogbonna (ogbonna@unfpa.org)
Benjamin SC Uzochukwu (bscuzochukwu@yahoo.com)
Agathe Lawson (alawson@unfpa.org)
Bannet Ndyanabangi (ndyabangi@unfpa.org)

Version: 7 Date: 15 January 2013

Author's response to reviews:

We are grateful to the reviewer for the comments. Our point-by-point responses to the comments are:

Reviewer

1. Table 7: it is ok that the denominator is the total numbers of respondents in each quintile, but the authors should note in the text that the percentages are dependent on the method mix within each quintile.

We agree with the reviewer and have included the text in table 7

2. The third paragraph from the end, under Discussion: Please remove the sentence "This explanation has been added to the discussion."

The reviewer is correct and the sentence has been removed.

3. The sentence before that "Strategies to promote..." would perhaps better go with the previous paragraph when the authors talked about the poorer SES not being aware of policies designed to increase their access to services.

We have removed the sentence and placed it on the previous paragraph as recommended.

Editorial:

We have added sections on competing interests and authors’ contributions.